Author's response to reviews

Title: Seizures as the first manifestation of chromosome 22q11.2 deletion syndrome in a 40-year old man: a case report

Authors:

Adriano R Tonelli (Adriano.Tonelli@medicine.ufl.edu)
Kalyan Kosuri (Kalyan.Kosuri@ht.msu.edu)
Sainan Wei (weisaina@msu.edu)
Davoren Chick (Davoren.Chick@hc.msu.edu)

Version: 5 Date: 25 November 2007

Author's response to reviews:

Dear BioMed Central Editorial Production Team:
I made the changes requested.
Thank you.
Sincerely,
Adriano Tonelli.

Formatting changes requested
-----------------------------

Country - Please remember to include the country in the affiliation details.
I included the country in the affiliation details and contact information.

Please place the Acknowledgements section after the authors' contributions.
I reorganized this heading.

Please create a new section titled 'Consent' and place this above the Acknowledgements, the text from the acknowledgements should then be moved accordingly.
I created a new section called "Consent" and I moved it above the Acknowledgements section.

References - The reference list should contain all authors' names, if there are 30 or below. The term 'et al.' should not be used.
I included all authors in the reference list (as it in PubMed).